Author's response to reviews

**Title:** Assessing Methods for Measurement of Clinical Outcomes and Quality of Care in Primary Care Practices

**Authors:**

Michael E Green (michael.green@dfm.queensu.ca)
William Hogg (whogg@uottawa.ca)
Colleen Savage (savagec@queensu.ca)
Sharon Johnston (SSJohnston@bruyere.org)
Grant Russell (grant.russell@monash.edu)
R. Liisa Jaakkimainen (liisa.jaakkimainen@ices.on.ca)
Richard H Glazier (rick.glazier@ices.on.ca)
Jan Barnsley (jan.barnsley@utoronto.ca)
Richard V Birtwhistle (richard.birtwhistle@cspc.queensu.ca)

**Version:** 5  **Date:** 27 June 2012

**Author's response to reviews:**

See above, no additional comments. Resubmitted revised format of one table as symbols were not converting properly.